Author's response to reviews

Title: Detection of a novel avian influenza A (H7N9) virus in humans by multiplex one-step real-time RT-PCR assay

Authors:

Jian Fan (3319fanjian@163.com)
David Cui (davidcui101@163.com)
Siu Ying Lau (sylau926@hkucc.hku.hk)
Guo Liang Xie (ytxiegouliang@163.com)
Xi Chao Guo (guoxichao926667@126.com)
Shu Fa Zheng (zsfzheng@163.com)
Xiao Feng Huang (stevehxf@hku.hk)
Shi Gui Yang (ysgcdc@163.com)
Xian Zhi Yang (yayayang1984@sina.com)
Zhao Xia Huo (huozhaoxia8253@163.com)
Fei Yu (yf0101@live.cn)
Jian Zhou Lou (justljz@gmail.com)
Li Tian (grapetian@163.com)
Xue Fen Li (zylixf@126.com)
Yue Jiao Dong (candy242313@163.com)
Qiao Yun Zhu (akuzhu@163.com)
Yu Chen (chenyu_zy@163.com)

Version: 4
Date: 24 July 2014

Author's response to reviews:

Dear Neil,

Thank you for your good comments. This is a carefully revised cover letter on reviews' comments!

We expect that the revised manuscript can be considered for acceptance.

Best regards,

David Cui